Author's response to reviews

Title: Preventing childhood obesity during infancy in UK primary care: a mixed-methods study of HCPs' knowledge, beliefs and practice

Authors:

Sarah A Redsell (Sarah.Redsell@nottingham.ac.uk)
Philippa J Atkinson (p.j.atkinson@bopenworld.com)
Dilip Nathan (dilip.nathan@nhs.net)
Aloysius N Siriwardena (nsiriwardena@lincoln.ac.uk)
Judy A Swift (judy.swift@nottingham.ac.uk)
Cris Glazebrook (cris.glazebrook@nottingham.ac.uk)

Version: 6 Date: 21 June 2011

Author's response to reviews:

Hi there

I have changed Cris' details on the manuscript. I am sorry but I am really confused about what you are asking me to do about Table 3. I know it is not a citation but an instruction to have the table put near to that text when it goes for final publication. It states quite clearly on page 10 of the version of the manuscript "Insert Table 3" at the end of the quantitative results before the section that states "phase 2 interviews". I have put a comment on it to show you where it is written. Do you want me to move the table there? I am happy for you to move it if that works for you - but it would sit best near the quantitative results that it relates to.

Thank you

Sarah